Author's response to reviews

Title: Lack of adherence to hypertension treatment guidelines among GP's in southern Sweden - a case report based survey

Authors:

Rickard W Ekesbo (rickard.ekesbo@skane.se)
Patrik Midlöv (patrik.midlov@med.luse)
Sofia Gerward (Sofia.gerward@med.lu.se)
Kristin Persson (perssonkristin@hotmail.com)
Christina Nerbrand (christina.nerbrand@med.lu.se)
Lennart Johansson (lennartkarin@telia.com)

Version: 5 Date: 14 March 2012

Author's response to reviews: see over
Dear Sir,

We again thank you for the opportunity to resubmit our paper and for the possibility to improve the manuscript.

Changes have been made according to the opinions made clear in your reply:

We have omitted the sentence believed to be unclear as well as the reference.

We have also rewritten the last sentence of the abstract as there may be alternative reasons explaining the outcome of the study. Although we find the issue of assessment of the GPs as the most plausible explanation we, as suggested, recognize this may not be the only option.

Hoping you find the manuscript improved sufficiently for publication, we remain

Yours sincerely,